\documentclass[a4paper,12pt]{revtex4}
\renewcommand{\baselinestretch}{1.3}
\usepackage{lscape,graphicx}
\usepackage{amsmath}
\usepackage{subfigure}
\usepackage[bf]{caption}

\setlength{\captionmargin}{10pt}

\setlength{\subfigcapskip}{1pt}
\setlength{\subfigtopskip}{1pt}
\setlength{\subfigbottomskip}{1pt}


\def\diadic#1{\overline{\overline{#1}}} 
\def\ni{\noindent}
\def\sig{\sigma}
\def\br{\vec{r}}
\def\bx{\vec{x}}
\def\bg{\vec{g}}
\def\bs{\vec{s}}
\def\bt{\vec{t}}
\def\bu{\vec{u}}
\def\bn{\vec{n}}
\def\ba{\vec{a}}
\def\bb{\vec{b}}
\def\bc{\vec{c}}
\def\bR{\vec{R}}
\def\bA{\vec{A}}
\def\bB{\vec{B}}
\def\bC{\vec{C}}
\def\bF{\vec{F}}
\def\bS{\vec{S}}
\def\bT{\vec{T}}
\def\bff{\vec{f}}
\def\bFF{\vec{F}}
\def\brho{\vec{\rho}}
\def\bomega{\vec{\omega}}
\def\bnabla{\vec{\nabla}}
%\def\hA{{\hat A}}
%\def\hB{{\hat B}}
%\def\hC{{\hat C}}
%\def\hp{{\hat p}}
%\def\hP{{\hat P}}
%\def\hG{{\hat G}}
%\def\hQ{{\hat Q}}
%\def\hq{{\hat q}}
%\def\hR{{\hat R}}
%\def\hS{{\hat S}}
%\def\hI{{\hat I}}
%\def\hsig{{\hat \sigma}}
%\def\heps{{\hat \epsilon}}
%\def\hdelta{{\hat \delta}}
%\def\hPi{{\hat \Pi}}
\def\hA{{\diadic{A}}}
\def\hB{{\diadic{B}}}
\def\hC{{\diadic{C}}}
\def\hD{{\diadic{D}}}
\def\he{{\diadic{e}}}
\def\hf{{\diadic{f}}}
\def\hG{{\diadic{G}}}
\def\hH{{\diadic{H}}}
\def\hK{{\diadic{K}}}
\def\hP{{\diadic{P}}}
\def\hQ{{\diadic{Q}}}
\def\hM{{\diadic{M}}}
\def\hN{{\diadic{N}}}
\def\hR{{\diadic{R}}}
\def\hS{{\diadic{S}}}
\def\hsig{{\diadic{\sigma}}}
\def\heps{{\diadic{\epsilon}}}
\def\hdelta{{\diadic{\delta}}}
\def\hPi{{\diadic{\Pi}}}
\def\hpi{{\diadic{\pi}}}
\def\eps{{\epsilon}}
\def\lX{\lambda X} 
\def\cA{{\cal A}}
\def\cL{{\cal L}}
\def\cW{{\cal W}}
\def\cP{{\cal P}}
\def\cO{{\cal O}}
\def\cF{{\cal F}}
\def\cD{{\cal D}}
\def\cH{{\cal H}}
\def\bcF{\vec{{\cal F}}}  
\def\halfspace{\hskip0.4cm} 
 
\def\halfspace{\hskip0.4cm} 
\def\parspace{\vskip0.4cm \ni} 
 

\begin{document}
\begin{thebibliography}{10}

\bibitem{Edetal} Edwards S. F., IMA Bulletin {\bf 25}, 94 (1989);
Edwards S. F. and Oakeshott  R. B., Physica D {\bf 38}, 88 (1989); 
Edwards S. F. and Oakeshott R. B., Physica A {\bf 157}, 1080 (1989);
Mehta A. and Edwards S. F., \textit{Physica} A {\bf 157}, 1091 (1989); 
Edwards S. F. and Grinev D. V., Phys. Rev. E {\bf 58}, 4758 (1998).
\bibitem{Chicago} Knight J.B., Fandrich C.G., Lau C.N., Jaeger H.M. and Nagel S.R., Phys. Rev. E {\bf 51}, 3957 (1995); 
Nowak E.R., Knight J.B., Povinelli M.L., Jaeger H.M. and Nagel S.R., Powder Technol. {bf 94}, 79 (1997); 
Nowak E.R., Knight J.B., Ben-Naim E., Jaeger H.M. and Nagel S.R., Phys. Rev. E {\bf 57}, 1971 (1998).
\bibitem{PhBi02} Philippe P. and Bideau D., Europhys. Lett. {\bf 60}, 677 (2002).
\bibitem{EntropySupport} Metzger P. T. and Donahue C. M., Phys. Rev. Lett. {\bf 94}, 148001 (2005); 
Makse H. A. and Kurchan J., Nature {\bf 415}, 614 (2002);
Barrat A., Kurchan, J., Loreto V., and Sellitto M., Phys. Rev. Lett. {\bf 85}, 5034 (2000);
Ono I. K., O'Hern C. S., Durian D. J., Langer S. A., Liu A. J. and Nagel S. R., Phys. Rev. Lett. {\bf 89}, 095703 (2002); 
Fierro A., Nicodemi M. and Coniglio A., Europhys. Lett., {\bf 59}, 642 (2002); 
Coniglio A., Fierro A., Nicodemi M., Ciamarra M. P. and Tarzia M., J. Phys.: Condens. Matter {\bf 17}, S2557 (2005).
\bibitem{BlEd03} Blumenfeld R. and Edwards S. F., Phys. Rev. Lett. {\bf 90}, 114303 (2003).
\bibitem{BlEd06} Blumenfeld R. and Edwards S. F., Euro. Phys. J., E {\bf 19}, 23 (2006).
\bibitem{QuadCaveat} Note that the quadrons may not tessellate space perfectly when cells are extremely non-convex, as discussed in \cite{Ci07}\cite{BlEd07}. Such cell may occur only in the presence of body forces, otherwise they cannot be in mechanical equilibrium. Even so, the probability of such cells is quite low and these cases are disregarded here.
\bibitem{Ci07} Ciamarra M. P., Comment on ``Granular entropy: Explicit calculations for planar assemblies'', Phys. Rev. Lett., in print (2007);
Blumenfeld R. and Edwards S. F., reply to comment on ``Granular entropy: Explicit calculations for planar assemblies'', Phys. Rev. Lett., in print (2007).
\bibitem{Bl07} Blumenfeld R., New J. Phys. {\bf 9} (2007) 160.
\bibitem{Euler} See e.g. Coxeterc H.M.S., Regular Polytopes (Dover, New York, 1973).
\bibitem{BlKi07} Blumenfeld R. and King P. R., Entropy-mediated structure-permeability relations in skeletal porous materials Water Resources Res., submitted.
\bibitem{BaBl02} Ball R. C. and Blumenfeld R., Phys. Rev. Lett. {\bf 88}, 115505 (2002).
\bibitem{Fretal07} Frenkel G., Blumenfeld R., Grof Z., King P. R., The structure, entropy and statistics of 2D granular systems, Phys. Rev. E., submitted.
\bibitem{GrRy} Gradshteyn I. S. and Ryzhik I. M., {\it Tables of Integrals, series, and Products}, (Academic Press, San Diego 1979).
\bibitem{BaPUNA} Ball R. C., in {\it Structures and Dynamics of Materials in the Mesoscopic Domain}, eds. M. Lal, R. A. Mashelkar, B. D. Kulkarni, and V. M. Naik (Imperial College Press, London 1999) 
\bibitem{Doetal} Donev A., Stillinger F. H., Chaikin P. M. and Torquato S., Phys. Rev. Lett., {\bf 92}, 255506, (2004).
\bibitem{AsWe00} Aste T. and Weaire D., {\it The Pursuit of Perfect Packing} (Institute of Physics, Bristol, 2000)
\bibitem{Kretal03} Kraynik A. M., Reinelt D. A. and van Swol F., Phys. rev. E {\bf 67}, 031403 (2003).
\bibitem{Ma46} Matzke E. B., Am. J. Botany {\bf 33}, 58 (1946). 
\bibitem{BlEd08} Blumenfeld R. and Edwards S. F., Compactivity and angoricity: granular stress statistics, in preparation.
\bibitem{Bl0408} Blumenfeld R., Phys. Rev. Lett. {\bf 93}, 108301 (2004). 
\bibitem{BlIMA} Blumenfeld R., in IMA Volume in Mathematics and its Applications {\bf 141} on {\it Modeling of Soft Matter}, eds. Maria-Carme T. Calderer and Eugene M. Terentjev, pp 235 (Springer-Verlag 2005).
\bibitem{Edinburgh} Bouchaud J. P., Cates M. E. and Claudin P. J., J. Phys. II (France) {\bf 5}, 639 (1995); 
Wittmer J. P., Claudin P., Cates M. E. and Bouchaud J. P., Nature {\bf 382}, 336 (1996); 
Cates M. E., Wittmer J. P., Bouchaud J. P. and Claudin P., Phys. Rev. Lett. {\bf 81}, 1841 (1998).
\bibitem{Geetal07} Gerritsen M., Kreiss G and Blumenfeld R., Stress chain solutions in two-dimensional isostatic granular systems: fabric-dependent paths, leakage and branching, Phys. rev. Lett., submitted.
\bibitem{ForceChainsExp} Liu C., Nagel S. R., Schecter D. A., Coppersmith S. N., Majumdar S., Narayan O. and Witten T. A., Science {\bf 269}, 513 (1995); 
Vanel L., Howell D., Clark D., Behringer R. P. and Clement E., Phys. Rev. {\bf E 60} R5040 (1999).
\bibitem{ForceChainsNum} Liffman K., Chan D. Y. and Hughes B. D., Powder Technol. {\bf 72}, 225 (1992); 
Melrose J. R. and Ball R. C., Europhys. Lett. {\bf 32}, 535 (1995).
\bibitem{ForceDistEmpirical} Lovoll, Maloy K. J., and Flekkoy E. G., Phys. Rev. {\bf E} , 5872 (1999); 
Corwin E. I., Jaeger H. M. and Nagel S. R., Nature {\bf 435}, 1075 (2005); 
Coppersmith S. N., Liu C. h., Majumdar S., Narayan O. and Witten T. A., Phys. Rev. {\bf E} 53, 4673 (1996); 
Rottler J. and Robbins M. O., Phys. Rev. Lett. {\bf 89}, 195501 (2002); Majmudar T. S. and Behringer R. P., Nature {\bf 435}, 1079 (2005).
\bibitem{Ed_ForceDist} Edwards S. F. and Grinev D. V., Granular Matter {\bf 4}, 147 (2003). 
\bibitem{EdBl07} Edwards S. F. and Blumenfeld R., in {\it Physics of Granular Materials}, ed. A. Mehta (Cambridge University Press, Cambridge 2007). 

\end{thebibliography}
 
 
\title{On the basis for entropic characterization of granular materials and applications}
\author{Raphael Blumenfeld} 
 
\affiliation{Earth Science and Engineering, Imperial College London, London SW7 2AZ, UK \\ and \\ 
Biological and Soft Systems, Cavendish Laboratory, JJ Thomson Avenue, Cambridge CB3 0HE, UK} 


\begin{abstract}
\begin{center}
{\bf Abstract}
\end{center}
This paper reviews recent development in entropic characterization of granular materials and derived several results. The use of the formalism is illustrated for the derivation of volume-based quantities throat size distribution. Finally, I discuss the relations between the entropic formalism and stress transmission in granular systems.  
\end{abstract}

\maketitle 


\section{Introduction: the entropic formalism} 

\ni The introduction of statistical mechanical methods to analyse assemblies of granular systems \cite{Edetal} has led to the development of new concepts in the field. 
Most notable are the concepts of compactivity, the analogue of temperature, and of a volume function, the analogue of the Hamiltonian.
The approach is based on a description of the entropy of the granular structure, namely, the statistics of configurations that a collection of grains can assume, given that they are confined to a container of a given volume $V$. This approach, developed originally by Edwards and collaborators \cite{Edetal}, has been prompted by experimental observations \cite{Chicago}\cite{PhBi02} of a reversible behaviour of post-vibrated granular beds (albeit with intriguing irreversible precursors). The reversibility, and more importantly the reproducibility of measurements of bulk properties, suggest the existence of an ensemble of equilibrium-like configuration states. Further support for the approach has come from numerical simulations \cite{EntropySupport}. 

\ni The formalism has many parallels with conventional statistical mechanics of thermodynamic systems. A central concept is a partition function
\begin{equation}
Z = \int e^{-\cW\left(\{q\}\right)/X} \Theta\left(\{q\}\right) D\{q\} \equiv e^{-Y/X}\ ,
\label{eq:Ai}
\end{equation}
that depends on: 
$\{q\}$ - a complete set of degrees of freedom (DOF); 
$\cW$ - a volume function that is the analogue of the Hamiltonian in thermodynamic systems and which depends on the DOF; 
$\Theta\left(\{q\}\right)$ - a probability density that describes the statistics of the DOF and imposes the constraint that the structure remain connected. Once these constraints, embodied in the form of $\delta$-functions, are satisfied, the function $\Theta$ can be regarded as the conventional density of states. 
The scalar $X\equiv 1/\beta$, named compactivity \cite{Edetal}, is the analogue of the temperature. The analogue of the free energy, named the free porosity, is $Y=-\ln Z / \beta$ \cite{BlEd03}\cite{BlEd06}. 
In terms of these, the configurational entropy is $S=\beta^2\partial Y / \partial\beta$ 
and the mean porosity is 
\begin{equation}
V = \frac{1}{Z}\int \cW(\{q\}) e^{-\beta \cW\left(\{q\}\right)} \Theta(\{q\}) D\{q\}
\label{eq:Aii}
\end{equation}
The entropy $S$ and the mean porosity $V$ are related through 
\begin{equation}
V = Y + X S = \partial(\ln{Z})/\partial(\beta) \ .
\label{eq:Aiii}
\end{equation}
Using this formalism, many other parallels can be, and have been, made between thermodynamic systems and granular systems.

\ni In spite of experimental and numerical testimonials, the general applicability of this approach has been controversial. In particular, it has not been clear how the idea of volumetric entropy can be used to understand macroscopic properties of granular systems. Nevertheless, the prospect of harnessing the power of statistical mechanical methods to the difficult problem of granular systems has been very appealing. 

\ni In addition to lingering skepticism, a significant hurdle to a regular use of the formalism has been the lack of a suitable explicit volume function $\cW$ that is both rigorously additive when summed over all grains and convenient for analytical and computational purposes. As a result, several approximations have been used in the literature, leading to model-sensitive results. The form of the volume function is significant both because it is the vehicle for the derivation of explicit estimates of structural properties and because it identifies the phase space that defines the structure of granular systems.
This problem has been resolved recently both in 2D \cite{BlEd03} and in 3D \cite{BlEd06}, with the introduction of a new partition of he granular space. In this description, grains are replaced by representative polyhedra (polygons in 2D), constructed from the contact network of the original grains. The grain polyhedra surround cell polyhedra that represent the pores of the original structure. This description makes it possible to tessellate the space by topologically identical units, called quadrons \cite{BlEd06}. The details of the constructions and the tessellations are explained in figures 1 (2D) and 2 (3D) and their captions. In terms of these, a volume function can be identified as a sum over quadron volumes \cite{QuadCaveat}
\begin{equation}
W = \sum_q V_q \ .
\label{eq:Aiv}
\end{equation}
The quadrons are the fundamental volume elements. A $z$-coordinated grain in 3D (2D) can be regarded as a composite of several quadrons - $z$ in 2D and $6(z-2)$ in 3D, a picture reminiscent of quarks in elementary particles.
 
\begin{figure}
\includegraphics[width=2.5in]{LectureNotesAusFig1.eps}
\caption{The geometric construction around grain $g$ in 2D. The vectors $\br^{cg}$ connect contact points clockwise around each grain anticlockwise around each void. The vectors $\bR^{cg}$ connect from grain centres to loop centres. The shaded quadrilateral is the quadron associated with $cg$.}
\label{fig:1}
\end{figure}

\ni This construction resolves the problems of disentangling the geometrical correlations and identifying the DOF that describe the structure \cite{BlEd03}\cite{BlEd06}.
The polyhedra are defined by their edge vectors $\br$. Structural correlations arise from irreducible loops, each giving a dependent vector. The number of these loops is straightforward to calculate in 2D, using Euler's relation \cite{Euler}, but not in 3D. These calculations are shown below. 

\ni Another advantage of this description is that it allows a local characterization of the structure by a fabric tensor, as described in \cite{BlEd03} (2D) and in \cite{BlEd06}\cite{BlKi07} (3D). The fabric tensors are useful both for quantifying the quadron volumes and for modelling stress transmission in granular assemblies \cite{BaBl02}. As such, these tensors are natural descriptors of granular systems. 

\ni In the remainder of the paper I illustrate the use of the entropic formalism for extraction of useful information, I clarify the relations between the statistics of the structure and the stresses that keep a granular system in mechanical equilibrium, and I finish with some considerations concerning the basis of the entire approach and its possible extensions.

\section{Calculations of volume-based structural properties}

\ni{{\it Two dimensions}}
\smallskip

\ni Let $V=\cW(\{q\})$ be the volume function of a system of $N$ grains, described by (\ref{eq:Aiv}), which depends on a set of structural DOF $\{q\}$. The partition function is 
\begin{equation}
Z = \int e^{-\beta\cW(\{q\})} \cP(\{q\}) D\{q\} \ ,
\label{eq:Ci}
\end{equation}
where $\cP(\{q\})$ is the density of states, i.e. the probability density of occurrence of particular configurations of $\{q\}$. 
Given a mean coordination number $\bar{z}$, there are $\bar{z}N/2$ contacts and $\bar{z}N$ vectors $\br$. On using Euler relation, this gives $\bar{z}/2$ irreducible loops, leaving $\bar{z}N$ DOF. Interestingly, this is also the number of quadrons and one can use the quadron volumes, $V_q$, to span the phase space.

\ni Absence of data for $\cP(\{q\})$ led to several approximations \cite{BlEd03}\cite{BlEd06}, but a recent study of it \cite{Fretal07} has revealed complex structure. 
In foam-like structure $\cP(\{q\})$ is fitted well by a gamma distribution, 
\begin{equation}
\cP(V_q) =  \frac{b^a V_q^{a-1}}{\gamma(a,bV_{max})-\gamma(a,bV_{min})}e^{-bV_q}  \ \ \ ; \ \ \ V_{min} \leq V_q \leq V_{max} \ ,
\label{eq:Ciii}
\end{equation}
where $V_{max} \gg 1/b$, $b$ is the inverse of a typical quadron volume and $3<a<4$ and $\gamma(a,A)$ is the incomplete gamma function \cite{GrRy}. 
An `ideal quadron-gas approximation' of uncorrelated quadrons gives 
\begin{equation}
Z =  \left[ C \int_{V_{min}}^{V_{max}}  V_q^{a-1} e^{-\beta V_q} dV_q \right]^{\bar{z}N} = \left\{
C \frac{\gamma\left[a,(b+\beta)V_{max}\right]-\gamma\left[a,(b+\beta)V_{min}\right]}{\left( 1 + \frac{\beta}{b}\right)^{a}} 
\right\}^{\bar{z}N} \ ,
\label{eq:Civ}
\end{equation}
The mean volume and volume fluctuations are, respectively,
\begin{equation}
\langle V \rangle =  \bar{z}N \frac{f_1(\beta)}{(b+\beta)} \quad {\rm and} \quad 
\langle V^2 \rangle - \langle V \rangle^2 =  \bar{z}N \frac{f_2(\beta) - f_1^2(\beta)}{(b+\beta)^2} \ ,
\label{eq:Cv}
\end{equation}
where
\begin{equation}
f_n(\beta) \equiv \frac{\gamma\left[a+n,(b+\beta)V_{max}\right]-\gamma\left[a+n,(b+\beta)V_{min}\right]}
{\gamma\left[a,(b+\beta)V_{max}\right]-\gamma\left[a,(b+\beta)V_{min}\right]} \ .
\end{equation}

\begin{figure} 
\includegraphics[width=4in]{LectureNotesAusFig2.eps}
\caption{ 
The polyhedral representation in 3D. The edge vectors $\br^{cg}$ connect contact points around grains and the construction of a 3D quadron is shown in stages. The quadrons are non-convex octahedra that tessellate the space perfectly. Grain volumes are composites of these quadrons.}
\label{fig:2} 
\end{figure}

\ni{{\it Three dimensions}}
\smallskip

\ni Euler relation is insufficient to determine the number of DOF in 3D. Let us focus on foam-like structures, where every grain has four contacts and is represented by a tetrahedron \cite{BlEd06}. 
Seen from within a void, the `surface' of a cell is made of the triangular faces of the $n_g^c$ grains surrounding it, of the contacts between the triangles and of the $n_f^c$ faces that the triangles enclose. 
The latter are the `throats' that connect to neighbour cells. 
There are $3n_g^c/2$ contacts and $3n_g^c$ edges on the surface, which, on using Euler relation, gives that a cell has $n_f^c = 2 + n_g^c/2$ throats.
Summing over all cells, remembering that every throat is shared between two cells, gives that there are $N_f = N + N_c$ throats, or $2(1+N/N_c)$ throats per cell. 

\ni Every polyhedron edge is a 3D vector. The polyhedron of a $z$-coordinated grain has $2(z-2)$ triangular faces and $3(z-2)$ edge vectors. The inter-dependency of all the edge vectors is due to geometric correlation. As in 2D, every irreducible loop obviates one vector. In foam-like structures, three of the six edge vectors of every tetrhedron are dependent. Additionally, each of the $N+N_c$ throats introduces a dependent vector. Therefore, there are in total $6N-3N-N_f=2N-N_c$ independent vectors, or 
\begin{equation}
\frac{N_{dof}}{N} = \frac{3(2N-N_c)}{N} = 3\left(2-\frac{N_c}{N}\right) 
\label{Ndof}
\end{equation} 
DOF per grain. This value is significantly lower than the number of quadrons per grain.

\ni The ratio $N_c/N$ is a key quantity. For infinitely rough convex particles, it can be bounded by using the dual structure. 
In this structure, the duals of the grain polyhedra are the cell polyhedra and visa versa. The contact surfaces between the dual grains are the original throats. 
In 2D the dual of a statically determinate such structure is also statically determinate, but for frictionless grains, and it is plausible that the same applies in 3D. 
In general, the larger is $z$ in the original system, the smaller the mean coordination number of the dual structure and visa versa. 
The lowest value that $\bar{z}$ can assume is determined by the condition of mechanical equilibrium for infinitely rough grains, $\bar{z}\geq 4$ \cite{BaPUNA}. 
The lowest mean coordination number for frictionless non-spherical grains is $12$ \cite{BaPUNA}.  
This corresponds to the number of faces per cell in the original structure, $z_c=2(N+N_c)/N_c$, hence $N_{c}/N \leq 1/5$. 

\ni The opposite bound is obtained as follows. 
The highest packing density of identical smooth and rigid ellipsoids is achieved within a narrow range of aspect ratios where $\bar{z}=14$ \cite{Doetal}. Polydispersity is expected to reduce $\bar{z}$. 
This is supported numerically \cite{AsWe00}\cite{Kretal03} and experimentally \cite{Ma46}. 
For $z_c=14$ we get a low bound: $N_{c}/N \geq 1/6$. 
Combining the two bounds with relation (\ref{Ndof}), we get 
\begin{equation}
27/5 \leq N_{dof}/N \leq 11/2 
\label{Ndofbounds}
\end{equation} 
and the phase space is $N_{dof}=(5.45 \pm 0.05)N$-dimensional. Thus, to span the phase space we need a fraction of the quadrons that is narrowly bounded between $9/20$ and $11/24$. 
Choosing these to be uniformly distributed in space, their low spatial density reduces the correlations between them. This makes it possible to use the ideal quadron gas approximation to compute the partition function and volume-based structural properties.  

\section{Other structural properties}
\bigskip

\begin{figure} 
\includegraphics[width=3in]{LectureNotesAusFig3.eps} 
\caption{ 
A non-planar throat of the polyhedral structure is a non-planar throat.  The vector $\bR_{cc'}$ extends between the centroids of the cells $c$ and $c'$ which the throat connects. The effective area to flow is estimated as the projection of the polygon onto a plane perperndicular to $\bR_{cc'}$.}
\label{fig:3}  
\end{figure}

\ni Not all structural properties are convenient to compute with the quadrons as the DOF. For example, an important characteristic of the structure is the mean throat size, which is directly relevant to the permeability to fluid flow. The areas of the throats are difficult to express in terms of quadron volumes, but straightforward to express in terms of the vectors $\br$. From the considerations above, the number of DOF per throat is 
\begin{equation}
\frac{9}{2} \leq \frac{N_{dof}}{N_f} = \frac{3(2-N_c/N)}{1+N_c/N} \leq \frac{33}{7} \ .
\label{NdofPerThroat}
\end{equation} 
The above also gives that the mean number of edge vectors around a face is 
\begin{equation}
5 \leq \frac{6N}{N_f} = \frac{6}{1+N_c/N} \leq \frac{36}{7} 
\label{EdgesPerThroat}
\end{equation} 
and therefore the number of variables per throat (three times this value) is larger than the number of DOF per throat. The computation of effective throat sizes is as follows. A throat is a non-planar polygon made of $n$ triangles. The effective area to flow of a throat between cells $c$ and $c'$, whose centroids are $A$ and $A'$, can be estimated by the polygon's projection onto a plane perpendicular to the vector $\bR_{cc'}$ extending from $A$ to $A'$ (see figure 3). Letting the polygon corners be $\brho_i'$, $i=0,1,...,n-1$, $\br_i = \brho_i' -\brho_{i-1}'$, $\brho_i = \br_i \times \bR_{cc'}$ and placing the origin at $\brho_0'$, the area of the projected throat is
\begin{equation}
A_{throat} = \frac{1}{2} \sum_{i=1}^{n-2} \brho_i \times \sum_{j=i+1}^{n-1}\brho_j \ ,
\label{ThroatSize}
\end{equation} 
giving the effective throat size in terms of the edge vectors. The mean throat size is then
\begin{equation}
\langle A_{throat} \rangle = 
\frac{1}{Z N_f}\int \sum_{throats} A_{throat} e^{-\beta\cW(\{\br\})} \cP(\{\br\}) D\{\br\} \ .
\label{MeanThroatSize}
\end{equation} 
Computing similarly expectation values of higher powers of the sum in (\ref{ThroatSize}), we can construct the distribution of throat sizes to any required accuracy using techniques developed for the moment problem.

\ni The extension of this calculation for realistic throat size distribution, where the grain shape distribution is taken into consideration, is straightforward in the entropic formalism and will be described elsewhere. Other structural properties can be calculated similarly, e.g. the total surface area between the solid and the pore space, useful for estimating heat exchange. 

\section{The entropic formalism and mechanical stresses}
\bigskip

\ni The discussion about the statistics of granular systems in mechanical equilibrium is not complete without taking into consideration the mechanical stresses that keep them in such states. The incorporation of stresses explicitly into the entropic formalism is a recent project in our group \cite{BlEd08}. Of particular interest is stress transmission in isostatic materials. Why these idealized systems are useful to understanding general granular materials has been discussed in \cite{Bl0408}\cite{BlIMA}.  
Stresses in planar systems are governed by 
\begin{equation}
\partial_i \sig_{ij} = g_j \quad ; \quad 
p_{xx}\sig_{yy} + p_{yy}\sig_{xx} = 2p_{xy}\sig_{xy}
\label{eq:Sic}
\end{equation}
with $\sig_{xy}=\sig_{yx}$ and $\bg$ including external and body forces. The rightmost is a constitutive stress-structure relation, whose parameters $p_{ij}$ characterize local structure. Their values were initially proposed empirically \cite{Edinburgh} and statistically \cite{Edetal} and eventually they have been derived from first principles. That derivation also highlighted their geometric interpretation on the grain level \cite{BaBl02}. Equations (\ref{eq:Sic}) can be solved under simplifying assumptions \cite{Bl0408}\cite{Bl07} and they have been analysed rigorously recently \cite{Geetal07}. All analyses confirm that localized source loads give rise in such media to force chains, in agreement with experiments \cite{ForceChainsExp} and simulations \cite{ForceChainsNum}.
Here, I discuss the relations between these solutions and the entropic formalism. 

\ni The statistical and pure mechanical descriptions have developed largely independently, but they must be related.
Measurements of contact force magnitude with exponential-tailed distributions led to statistical-mechanical based explanations that are independent of the solutions of (\ref{eq:Sic}) \cite{ForceDistEmpirical} \cite{Ed_ForceDist}\cite{EdBl07}. Disregarding these solutions necessitated introduction of assumptions that weakened the models and led to much controversy.

\ni This point can be illustrated in jamming of slowly sheared granular systems. The microstructure changes continuously in response to forces until the systems jams. We can now solve for the stress field in the jammed state, but it is the very stress solution which affected the structural characteristics. 
Thus, a full statistical mechanical description must include both the structural information and the boundary loading. 

\ni It is important to recall here two important differences between granular and thermal systems. Measurements in the latter are normally done on time scales that allow the system sufficient time to explore a sufficiently large number of states to render the data typical. Granular systems are in principle quenched in a given state, especially for stress measurements. The only statistics involved here could be through ensembles of systems. Another significant difference is that eqs. (\ref{eq:Sic}) are hyperbolic, which is the reason for the chain solutions. 
This means that fluctuations in boundary force loads can be felt far from the load source. 
For example, consider a granular material pressed within a container by a flat plate of area  $S$ with a force $F$. 
Since the boundary of a granular pack is never flat then the plate presses on protruding grains differently than on their neighbours. 
These locally elevated forces act as localized load sources and give rise to chains. 
If the typical distance between such sources along the boundary is larger than the scale of resolution, or interest, then the particular distribution of chains is more significant than the mean pressure $F/S$. 
This suggests that: 
(i) care should be taken in the specification of the boundary loads and 
(ii) that one cannot ignore these fluctuations in the analysis of the statistics.

\ni To address this issue we expand the phase space of DOF to include an ensemble of all the possible loads \cite{BlEd08}\cite{EdBl07}. Consider an ensemble of all possible systems in mechanical equilibrium with given volume $V$ and total boundary stress $\hPi$. Divide this ensembles into two sub-ensembles. One, the $\Pi$-ensemble, consists of all possible configurations of the particles that occupy a given volume $V$ under a boundary loading of a given spatial distribution of force loads on the boundary. The other sub-ensemble, the V-ensemble, consists of all the possible realizations of boundary forces that add up to the boundary stress $\hPi$, such that the particular configuration of particles is kept fixed. Note that these are in fact all the stresses that are confined to within the so-called yield surface for this configuration. 

\ni The partition function can now be written as 
\begin{equation}
Z = \int e^{-\cW\left(\{q\},\{f\}\right)/X_0 - \cF\left(\{q\},\{f\}\right)/\left(VX_{ij}\right)} \Theta\left(\{q\},\{f\}\right) D\{q\}D\{f\} \ ,
\label{VPartition}
\end{equation}
where $\{f\}$ are the DOF describing the boundary loads, e.g. by the forces on every boundary grain, $\cF$ is the force moment (from which stresses are defined) and the function $\Theta$ is a product of $\delta$-functions requiring both that the particles are in contact and that Newton's equations are satisfied.
The variable $X_0$ is the compactivity, previously denoted by $X$, and $X_{ij}$ are the components of a tensorial analogue of temperature defined via derivatives of the entropy $S$ with respect to the boundary stresses
\begin{equation}
X_{ij} = \frac{\partial S}{\partial \Pi_{ij}} \ .
\label{Angoricity}
\end{equation}
The tensor $X_{ij}$ has been named angoricity \cite{EdBl07}. 

\ni The use of this formalism is currently explored in our group. In particular, we are looking into developing a Boltzmann equation, taking into consideration both the tensorial structural description and the newly discovered stress solutions into consideration.

\section*{Acknowledgements}
RB acknwledges the support of EPSRC grant no.GR/T28959/01.




\end{document}

